Author's response to reviews

Title: Nurse anaesthetists- and OR nurses` adherence to WHO's Surgical Safety Checklist -a grounded theory study

Authors:

Hilde V Waehle 1 (hilde.valen.waehle@helse-bergen.no)
Arvid S Haugen 1,2 (arvid.haugen@helse-bergen.no)
Eirik Softeland 1 (eirik.softeland@helse-bergen.no)
Esther Hjälmhult 3 (Esther.Hjaelmhult@hib.no)

Version: 4 Date: 10 July 2012

Author's response to reviews:

Dear BioMed Central,
Editorial Team
July 9th 2012

Please find our manuscript enclosed:
1608142719647704
«Adjusting team involvement: A grounded theory study of challenges in utilizing a surgical safety checklist as experienced by nurses in the operating room» which we would like you to consider for publication in Bio Medical Nursing.

As BMC Nursing is an open access, peer-reviewed journal, we consider it of great value to get our manuscript published in this journal.

We declare that we have no competing interest, and that the manuscript is not under consideration elsewhere. There is no financial support. In accordance with Norwegian law, approval by the Western Norway Regional Medical and Health Research Ethics Committee was not required.

Respectfully,
On behalf of the “Safe Surgery Study group”
Hilde Valen Wæhle
MSc, RNA

Correspondence address:
Hilde Valen Wæhle
Department of Anesthesia and Intensive care,
Haukeland University Hospital,
5021 BERGEN
Norway
E-mail: hilde.valen.wehle@helse-bergen.no
Telephone: +47 55 97 24 55 / +47 920 86 0 90